Author's response to reviews

Title: Adult asthma and traffic exposure at residential address, workplace address, and self-reported daily time outdoor in traffic: A two-stage case-control study.

Authors:

Anna Lindgren (anna.lindgren@med.lu.se)
Jonas Björk (jonas.bjork@med.lu.se)
Emilie Stroh (emilie.stroh@med.lu.se)
Kristina Jakobsson (kristina.jakobsson@med.lu.se)

Version: 6 Date: 18 November 2010

Author's response to reviews:

Cover Letter: All requested major and minor final changes have been done.

1. We have provided a title for Table 1.
2. All additional files are now mentioned in the text in numerical order and a section titled "Additional files" has been provided at the end of the manuscript.
3. "Objectives have been changed to 'Background' in the abstract.
4. All highlighting has been removed from the manuscript.
5. All vertical lines have been removed from the tables
6. The figures have now minimal white space around them.